Author’s response to reviews

Title: Associations between successful palliative cancer pathways and community nurse involvement

Authors:

Mette Asbjoern Neergaard (man@alm.au.dk)
Peter Vedsted (p.vedsted@alm.au.dk)
Frede Olesen (fo@alm.au.dk)
Ineta Sokolowski (ineta.s@alm.au.dk)
Anders Bonde Jensen (andjense@rm.dk)
Jens Sondergaard (JSoendergaard@health.sdu.dk)

Version: 16 Date: 3 December 2009

Author’s response to reviews: see over
Concerning MS: 1768922933258041
Associations between successful palliative cancer pathways and community nurse involvement

Dear BioMed Central Editorial Production Team

We thank you for your corrections, which we fully agree on. We found an error in Table 3: The numbers in column 2 did not match the data. Corrections have been made.

On behalf of the authors,
Yours Sincerely

Mette Asbjørn Neergaard
Speciality Registrar, PhD
The Palliative Specialist Team
Aarhus University Hospital
Noerrebrogade, DK-8000 Aarhus C
And
The Research Unit for General Practice
University of Aarhus
Bartholins Alle 2, DK-8000 Aarhus
Phone: +45 8942 6010
Email: man@alm.au.dk